Author's response to reviews

Title: A community pharmacy weight management programme: an evaluation of effectiveness.

Authors:

David S Morrison (david.morrison@glasgow.ac.uk)
Philip McLoone (philip.mcloone@glasgow.ac.uk)
Naomi Brosnahan (naomi.brosnahan@counterweight.org)
Louise McCombie (louise.mccombie@counterweight.org)
Andrea Smith (andrea.smith5@nhs.net)
Janie Gordon (janie.gordon@nhs.net)

Version: 4 Date: 25 September 2012

Author's response to reviews:

Dear Editor

I have amended the manuscript to include a statement on ethical committee approval and consent.

The study was an audit of a planned delivery of an existing intervention, and no new or untested treatment was being offered. Ethical approval was not necessary.

Patients gave consent to participation when they signed up to the Keep Well programme, however no personally identifiable data were collected or used for this particular study and written consent was not required.

Your sincerely

Philip McLoone